Author's response to reviews

Title: Screening of Crude Extracts of Six Medicinal Plants used in South-West Nigerian Unorthodox Medicine for Anti-Methicillin Resistant Staphylococcus aureus activity.

Authors:

Kabir’ O Akinyemi (akinyemiko@yahoo.com)
Olukayode Oladapo (oladapokayode2005@yahoo.com)
Okwara E Clement (okwaraclement@yahoo.com)
Christopher E Ibe (ibechris2002ng@yahoo.com)
Kehinde .A Fasure (fasureak@yahoo.com)

Version: 3 Date: 13 January 2005

Author's response to reviews:

The Editorial Team,

On behalf of other contributing authors, I hereby forward the revised version of the above titled article in line with the reviewer's comments.

1. The manuscript has been proof-read with the assistance of English competent Secretariat.

2. The grammatical and spelling mistakes have been rectified.

3. The terminology "MRSA" (Methicillin Resistant Staphylococcus aureus) has now been adopted for uniformity.

4. The format of table is now consistence.

5. The question raised on the 'immunomodulation' in the discussion has now been addressed.

6. Other suggestion raised by the reviewer has been adequately noted and addressed.